Author's response to reviews

Title: Protocol for a pragmatic randomised controlled trial to evaluate effects of a brief intervention for emergency department attendees who present with moderate and high levels of non-specific psychological distress

Authors:

Petra Lawrence (petra_lawrence@health.qld.gov.au)
Paul Fulbrook (paul.fulbrook@acu.edu.au)
Shawn Somerset (shawn.somerset@acu.edu.au)
Paula Schulz (paula.schulz@acu.edu.au)

Version: 3 Date: 5 May 2014

Author's response to reviews:

The latest revisions
1. Removal of Shawn Somerset and Paula Schulz as authors. These names now appear in the acknowledgements section.
2. The trial registration number now appears after the authors section (before the abstract). Also included is the full name of the trial registry and the date it was registered.
3. The funding source is in the acknowledgements section and is more detailed as the project itself is not funded, but PL's time is funded by an Australian Post graduate Award PhD scholarship.
4. In-kind funding is reported in the competing interests section.